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3. LOOKING AHEAD

XDC:

WEBSERVER: Evaluate performance issues with new web design. Need to consider 
alternate configurations with load­balanced servers and new DB machines with 
more memory.

ACRF SECURITY:  Improve security at ACRF Web sites containing sensitive 
information (See ECO­612 and ECO­563). SSL Certificates are in place. 
LDAP servers are running at BNL and ANL and the BNL server is getting 
any updates from the PeopleDB and is replicating those updates to the 
ANL server. Update:  Installed and configured new hardware for LDAP 
server at BNL. Next: Develop applications, e.g.: PeopleDB profile editor,  
IT VLAN and examine options for implementing a single sign-on system.

AMF Ancillary Facility Data for China: Make data available before STM.  Update:  
Completed the data for the first deployment site (Zhangye, Western China),  
Additional data for the second deployment (Xianghe) have been delivered to the  
XDC. Next: Complete metadata for data delivered.

WEB SYSTEM: Generate usage statistics for all web pages; analyze trend over ACRF 
whole period of existence.  

ASP FC DATA: Merge past ASP data into the FC metadata system.

5. NEW INSTRUMENTS AND DATA STREAMS

XDC:

DATASETS SUBMITTED TO THE FIELD CAMPAIGN BROWSER THROUGH THE      
XDC:   

CJC DATA­ Continued organizing data for the RHUBC­2 Experiment 
http://iop.archive.arm.gov/arm­iop/2009/cjc/rhubc­ii/

6. ACCOMPLISHMENTS

http://iop.archive.arm.gov/arm-iop/2009/cjc/rhubc-ii/


8. ISSUES 

XDC:

GOOGLE: The Google Mini (the replacement for the Google Search Appliance) has 
been installed.  It has been configured and is currently undergoing testing.  The 
plan is to place it into production next month, pending testing results (see BCR­
01591).

ASR WEBSITE:  Set up MySQL database for ASR WordPress.

ASR STM:  Finished the poster PDF upload form and administrator tools to accept, 
delete or upload PDF files.

NEWDESIGN WEBSITE:  Work continued on the php template for the XDC pages, the 
Publications Database Application, Conference Proceedings and the Research 
Highlights. 

CSPHOT & SUOMINET MENTORSHIP:  Continue to monitor Cimel issues. 
Reprocessing and updating level 2 data. Updating ingest for level 2, version 2 
Data.  Arranging the swap and calibration of instruments #98 and #252. Continue 
to review and consult on ARRA purchase (ECO­00765).  Details on individual 
Cimel issues are available in the IMMS report http://www.db.arm.gov/IMMS/.

SSM/I: SSM/I data were reprocessed to pick up data that was corrected byRemote 
Sensing Systems (RSS). RSS applied a correction to the F15 data which have 
problems due to interference from the RADCAL beacon. Entered DQPR 2378 
and associated DQR D100303.1 to let users know that the F15 data should be 
used with caution. The F16 data should be available in a few months.  Also 
entered BCR­01639 to end the companion quick­look image file collection.

GFS GRIB:  Updating the ingest to work with GRIB2 files and changes in variables. 
Reviewing DOD header.

PeopleDB:  Added all recent DB entries (since Oct. 2009) to the armall email list. 
Created an affiliation management page to display and update affiliation 
information.

FC METADATA:  The metadata for the ISDAC AAF Field Campaign are completed and 
awaiting final implementation.  Next on the list is completing the metadata for the 
China AMF.  A summer student is lined up for assistance with entering FC 

http://www.db.arm.gov/IMMS/


metadata.  A modification to the FC admin tool allows a pi­inst pair to be cloned, 
saving time.

TDMA:  The size distribution and hygroscopic ingests are complete.  We are awaiting 
documentation (waiting on TDMA­APS file integration processing, in addition to 
some quality flag issues) before they are officially announced.  The CCN data 
ingest has been completed.  Data are available at the Archive, with documentation 
available at the XDC.

Don has provided new sample integrated TDMA­APS data for the size 
distribution measurements.  This will require an update to the size distribution 
ingest, which will also result in a new data stream (hydroscopic ingest will remain 
the same.)  The new TDMA­APS data stream ingest is in development, and is 
undergoing error testing.  An error was found with duplicate time entries in the 
new TDMA­APS sample data, which is being addressed by Don Collins.  Waiting 
for new data (size distribution, hygroscopic, CNN and new TDMA/APS).  Work 
on the collection script is on hold.  See ECO­00587, BCR­01491.  

DQPR/PIFCARDQR: Work continues on requests for additions and bug fixes to the 
newly released DQPR2 system (see EWO­13212). 

FILE TRANSFER STATS: The XDC sent 21,528 files with a volume of ~530 GB this 
month.  


	DATASETS SUBMITTED TO THE FIELD CAMPAIGN BROWSER THROUGH THE XDC:

